Author’s response to reviews

Title: High genital prevalence of cutaneous human papillomavirus DNA on male genital skin: The HPV Infection in Men Study

Authors:

Laura Sichero (lsichero@gmail.com)
Christine M Pierce-Campbell (Christine.PierceCampbell@moffitt.org)
William Fulp (William.Fulp@moffitt.org)
Silvaneide Ferreira (silvaneide.ferreira@icesp.org.br)
João S Sobrinho (simao_j@yahoo.com.br)
Maria L Baggio (izamlbaggio@gmail.com)
Lenice Galan (lenicegalan@gmail.com)
Roberto C Carvalho (rjcssp@terra.com.br)
Eduardo Lazcano-Ponce (elazcano@insp.mx)
Anna R Giuliano (Anna.Giuliano@moffitt.org)
Luisa L Villa (luisapvilla@gmail.com)

Version: 3 Date: 26 November 2014

Author’s response to reviews: see over
November 26th, 2014

Dear On behalf of Nathaniel Nazreno
Journal Editorial Office
BioMed Central

Re: # 1446586435135677 - revised manuscript

Following up on your e-mail of November 26th, 2014 we had the opportunity to revise the aforementioned manuscript. Changes were incorporated in the manuscript as suggested.

Comments from the Editor:
"Please remove their additional file and include the names of the specific ethics committees that approved the study."

We included the information required in the text: “The IRD of the University of South Florida, Tampa, USA, the Instituto Mexicano del Seguro Social, and the Instituto de Salud Publica de Mexico, Mexico, and the Centro de Referência e Treinamento DST//Aids, Brazil, approved all study procedures, and all participants provided written informed consent.”

We thank you in advance for the opportunity of submitting this manuscript to BMC Infectious Disease and look forward to hearing from you

Sincerely yours,

Laura Sichero

Centro de Investigação Translacional em Oncologia – ICESP
Av. Dr. Arnaldo, 251, 8 andar
01246-000 Cerqueira César - São Paulo, SP
Phone: 55 11 3893 3010